Author’s response to reviews

Title: The prevalence of dementia in a Portuguese community sample: A 10/66 Dementia Research Group study

Authors:

Manuel Gonçalves-Pereira (gpereira@nms.unl.pt)
Ana Cardoso (amargaridapc@gmail.com)
Ana Verdelho (averdelho@medicina.ulisboa.pt)
Joaquim Alves da Silva (jalvesdasilva@nms.unl.pt)
Manuel Caldas de Almeida (malmeida@scmmora.pt)
Alexandra Fernandes (alexandrafernandesmgf@gmail.com)
Cátia Raminhos (csraminhos@gmail.com)
Cleusa Ferri (ferricleusa@gmail.com)
A. Matthew Prina (matthew.prina@kcl.ac.uk)
Martin Prince (martin.prince@kcl.ac.uk)
Miguel Xavier (migxavier@gmail.com)

Version: 2 Date: 10 Oct 2017

Author’s response to reviews:

Ref: Manuscript BGTC-D-17-00158-R1, submitted to BMC Geriatrics: The prevalence of dementia in a Portuguese community sample: A 10/66 Dementia Research Group study


Lisbon, 10th October 2017
Dear Handling Editor - BMC Geriatrics, Prof. Dr. Dr. M. Cristina Polidori,

Dear Anne Menard and Rien Ducot, BMC Geriatrics,

Thank you so much for your letter dated 9th October 2017. We have addressed the required points as you may see in the revised submission:

1) and 2): Thanks for the opportunity of providing a more detailed justification regarding the qualification of all of us to authorship. As requested, we briefly expanded the ‘Authors’ contributions’ section, also stating specifically (as we had also made clear in our previous cover letters) that all authors have read and approved the manuscript.

3) We have corrected Prof.’s Miguel Xavier email address in the manuscript.

4) The manuscript was uploaded unchanged (unless for points 1-3 above), with tables (also left unchanged) in a separate file.

We sincerely hope that the manuscript may now be found acceptable for publication in BMC Geriatrics.

Looking forward to your final decision,

Kind regards,

Manuel Gonçalves-Pereira